

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Hasan, Iftekhar; Song, Liang; Wachtel, Paul

Working Paper Institutional development and stock price synchronicity: Evidence from China

BOFIT Discussion Papers, No. 20/2013

Provided in Cooperation with: Bank of Finland, Helsinki

Suggested Citation: Hasan, Iftekhar; Song, Liang; Wachtel, Paul (2013) : Institutional development and stock price synchronicity: Evidence from China, BOFIT Discussion Papers, No. 20/2013, ISBN 978-952-6699-25-7, Bank of Finland, Institute for Economies in Transition (BOFIT), Helsinki, https://nbn-resolving.de/urn:nbn:fi:bof-201408072083

This Version is available at: https://hdl.handle.net/10419/212775

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

BOFIT Discussion Papers 20 • 2013

Iftekhar Hasan, Liang Song and Paul Wachtel

Institutional development and stock price synchronicity: Evidence from China



EUROJÄRJESTELMÄ EUROSYSTEMET Bank of Finland, BOFIT Institute for Economies in Transition BOFIT Discussion Papers Editor-in-Chief Laura Solanko

BOFIT Discussion Papers 20/2013 12.8.2013

Iftekhar Hasan, Liang Song and Paul Wachtel: Institutional development and stock price synchronicity: Evidence from China

ISBN 978-952-6699-25-7 ISSN 1456-5889 (online)

This paper can be downloaded without charge from http://www.bof.fi/bofit.

Suomen Pankki Helsinki 2013

Contents

Abs	stract		4
1	Intr	oduction	5
2	Hyp	ootheses and related literature	8
3	Data	a	11
	3.1	Sample	. 11
	3.2	Measurement of stock price synchronicity	11
	3.3	Measures of legal and political institutions	13
	3.4	Other variables	14
	3.5	Descriptive statistics	15
4	Bas	eline regression results	16
5	Mo	del extensions and robustness checks	19
	5.1	Alternative measures of institutional quality	19
	5.2	Threshold effects	20
	5.3	Alternative explanations	21
	5.4	Additional robustness checks	23
6	Con	clusions	.24
Ref	erenc	ces	25
Tab	les a	nd figures	. 28

Iftekhar Hasan, Liang Song and Paul Wachtel

Institutional development and stock price synchronicity: Evidence from China

Abstract

Better developed legal and political institutions result in greater availability of reliable firm-specific information. When stock prices reflect more firm-specific information there will be less stock price synchronicity. This paper traces the experience of China, an economy undergoing dramatic institutional change in the last 20 years with rich variation in experiences across provinces. We show that stock price synchronicity is lower when there is institutional development in terms of property rights protection and rule of law. Furthermore, we investigate the influence of political pluralism on synchronicity. A more pluralistic regime reduces uncertainty and opaqueness regarding government interventions and therefore increases the value of firm-specific information that reduces synchronicity.

JEL Classification Numbers: G14; G15; G24; G38

Keywords: Institutions; China; stock price synchronicity.

Liang Song, Michigan Technological University

Iftekhar Hasan, Fordham University and Bank of Finland, 5 Columbus Circle, 11th Floor, New York, NY 10019, United States. E-mail: <u>ihasan@fordham.edu</u>

Paul Wachtel, Stern School of Business, New York University

Acknowledgement: The authors are grateful to Diyue Guo for excellent research assistance, to the editor and the reviewers for their suggestions and Randall Morck and Thorsten Beck for their helpful comments. Earlier versions of this paper were presented at the conference on "Bridging the GAAP: Recent Advances in Finance and Accounting" at Hebrew University, Jerusalem, Israel, in July 2010, and at the "Conference on Chinese Capital Markets" at the New York University Center on US-China Relations, New York in May, 2011.

1 Introduction

The influence of institutional characteristics on the synchronicity of stock prices can be traced to Grossman and Stiglitz (1980), who argued that informed trading and informative pricing, which are central determinants of firm-specific stock variation, are determined by the costs and benefits of information collection. Following their rationale, we argue that poor legal institutions increase the cost of information collection and reduce investors' incentives to collect private information. This reduces the information content of stock prices and increases stock price synchronicity. Furthermore, we introduce a related argument regarding the openness of political institutions, namely, that there is greater availability of private information in more open political environments.

For the most part, the existing literature relates the quality of institutions to economic performance with cross country studies in which the heterogeneity of country specific characteristics makes it difficult to identify the effects even with panel data (Acemoglu et. al., 2005; Glaeser et. al., 2004; Wachtel, 2011). In this paper we focus on one country, China, where the pace of both legal and political development has varied across the provinces, and we relate these developments to the quality of information and, hence, synchronicity in equity markets. Rapid development over the last 20 years has both improved Chinese institutions and increased the variation among China's provinces in the level of legal and political development. Specifically, we use a panel of Chinese provinces to investigate the effect of institutions, both legal and political, on stock price synchronicity.

We show that the differences in stock price synchronicity are in part due to the variation among the provinces in institutional characteristics regarding property rights protection and the extent of the rule of law, as well as the degree of political pluralism. That is, stock price synchronicity is lower (i.e., the information content of stock prices is higher) in provinces with a better legal environment and greater political openness.

A priori, there are two reasons why poor institutions can lead to less collection of private information. First, in provinces where property protection for investors is weak, the risk of expropriation by corporate insiders could make firm-specific information less valuable, and informed risk arbitrage is unattractive. Second, even good regulations regarding information disclosure are often not fully enforced when there is poor rule of law. The entrenched managers are able to withhold relevant information to cover their own self-

serving behavior.¹ The argument regarding political pluralism is similar. A more pluralistic political structure reduces the uncertainty and opacity of political decision making because the dominant Communist party has to contend with non-party members and is subject to increased monitoring from them. Thus, greater political pluralism in a province will reduce the costs that result from political risk in collecting private information, which, in turn, decreases synchronicity.

The cross country literature on synchronicity has shown that firm-specific stock return variation is high or stock price synchronicity is low in countries with more developed institutions because reliable firm specific information is more readily available (Morck, Yeung and Yu, 2000). Furthermore, countries where investors have more firm-specific information will also have greater efficiency of capital allocation and lower costs of capital (Durnev, Morck and Yeung, 2004; Wurgler, 2000). However, a potential drawback of the cross country studies is that firms operating in different national environments are also affected by other country specific characteristics, such as diversity of historical experience. We are able to hold national characteristics constant by focusing on the province-level differences in institutional development across China.

The literature on synchronicity focuses on the effects of investor protection from the perspective of the legal and information environment with no discussion of the direct influence of the political environment (i.e., Jin and Myers, 2006; Fernandes and Ferreira, 2009). We suggest that the political environment may be equally important because politicians can change investor protections if they choose to do so (Pagano and Volpin, 2005). Political authorities can alter the legal environment with legislative changes or by changing the way laws are applied. An authoritarian political regime can enact new legislation or change the existing legal environment as it relates to investor protection. There is significant variation across China's provinces in the extent of influence of the Communist party and the diversity of the political environment. We show that differences across the provinces in political pluralism influence stock price synchronicity.

Stock price synchronicity among the Chinese provinces has been examined by Gul, Kim and Qiu (2009) in a study of ownership concentration and corporate governance. We adopt their approach to the measurement of synchronicity and apply it to a study of local institutions. The measures of legal and political development in China at the provin-

¹ For a discussion of these issues in emerging markets, see Chan and Hameed (2006) and Fan and Wong (2005).

cial level were first developed by Hasan, Wachtel and Zhou (2009), who use the data to explain differences in the growth rate across provinces. Data are available for 31 Chinese provinces for the period 1998-2007.

The possible endogeneity of synchronicity presents a major challenge which we attempt to solve by using GMM estimation. The results are often over identified, but when they are not we find that our results with OLS and GMM are largely the same. Furthermore, we present a number of tests to support the robustness of our results. For example, Morck, Yeung, and Yu (2000) find that the relationship between investor protection and stock price synchronicity is different in emerging markets as compared to developed countries. There are large differences in the level of development across China; we divide our sample between those provinces with above and below median GDP per capita. Our results are not driven by the differences between the two sub samples. We also explore some alternative explanations for the roles of institutions. For instance, Khanna and Thomas (2009) suggest that poor institutions may enhance the development of business groups, vertical integration, and operating diversification, which further increases stock price synchronicity. We examine the role of corporate governance by looking at the interaction between our institutional measures and firm characteristics, such as ownership that is related to better corporate governance. We find that better institutions have a more pronounced effect on stock price informativeness for firms with poorer corporate governance, as indicated by a higher proportion of government ownership or a lower proportion of foreign ownership. These results support our hypothesis that the negative relationship between institutional development and stock price synchronicity is due to poor institutional development that provides less investor protection. This increases the cost of information collection and reduces investors' incentives to collect private information, which further reduces the information content of stock prices and increases stock price synchronicity.

Our study contributes to the literature in several ways. First, as noted earlier, by using province level data from one country, we overcome the problem faced by all cross country studies regarding omitted country characteristics. We provide a unique focus on the effect of province-level investor protection in the single most important emerging market, China. Second, in addition to the effects of property rights and law enforcement, we also examine the effect of another important aspect of institutional development, political pluralism—a measure of the strength of democracy—on stock price informativeness, which has not been previously investigated. Third, China utilizes a quota system to select

7

companies from each province for listing or raising additional equity on a stock exchange. Du and Xu (2009) show that provinces in which listed firms have more firm-specific information incorporated into stock prices were rewarded with more stock quotas in the subsequent periods. Thus, our study shows that policies to improve institutions can also increase access to capital.

The remainder of the paper proceeds as follows. Section 2 discusses the specific hypotheses that we will test. Section 3 presents the calculation of synchronicity, our other measures, data sources, and descriptive statistics. Section 4 describes our empirical methodology and provides the baseline estimates of the relationship between institutional development and stock price informativeness. Section 5 provides extensions of the results, including an examination of the investment behavior of firms and our robustness tests. The final section concludes.

2 Hypotheses and related literature

China has followed an incremental approach to reforming its centrally planned economy into a market system, which has resulted in remarkably high growth rates for almost three decades (Prasad and Rajan, 2006). Rapid economic growth both demands and facilitates the development of legal institutions and the evolution of political systems (Williamson, 1996). There are a few empirical studies on the role of institutional development in China that show, for example, that the institutional environment matters for Chinese firms' reinvestment decisions and for fostering entrepreneurship (Cull and Xu, 2005; Djankov, Qian, Roland and Zhuravskaya, 2006).

An important feature of the evolution of the legal environment in China is that many of the laws adopted are enacted locally following national legislation; as such, their implementation varies from province to province (Krug and Hendrischke, 2003). Such variation is found for laws and regulations that ensure the protection of property rights (Hasan, Wachtel and Zhou, 2009).

Studies with cross country data have established the importance of intellectual property rights protection for economic growth (Gould and Gruben, 1996; Park and Ginarte, 1997). Hasan, Wachtel, and Zhou (2009) use the number of trademark applications in Chinese provinces as a proxy for the awareness of property rights and show that it is related to growth rates.

The link between property right protection and the information in stock prices has been discussed in Morck, Yeung, and Yu (2000). They argue that poor protection for property rights could discourage informed risk arbitrage because expropriation risk will make such behavior less valuable and increase the cost of collecting firm-specific private information. The resulting reduction in informed trading will impede the capitalization of firm-specific information into stock prices and decrease firm-specific return variation. Thus, they show that equity returns of firms in economies with poor country-level property rights exhibit relatively high synchronicity.

Our first hypothesis is: **H1:** *Stock price synchronicity is lower for firms in provinces with better property rights.*

Another notable feature of the evolution of the modern Chinese legal system was the spread of the rule of law, specifically the gradual development of institutional structures for the enforcement of rules and the settlement of disputes. For instance, in 1995, the State Compensation Law allowed citizens to sue the state (Burns, 1999). The functions of the Party and government in the People's Congress system were separated to strengthen the rule of law (Burns, 1999). In this process, many members of the legal profession, such as lawyers and judges, were rehabilitated, which significantly improved the quality of law enforcement (Hasan, Wachtel, and Zhou, 2009).

The effect of rule of law and related issues on firm-specific return variation were explored by Ball (2001) and Chan and Hameed (2006). Even when there are adequate laws on the disclosure of corporate information, they are often not fully enforced when there is poor rule of law. Entrenched managers might find it in their interests to withhold information, which increases the cost of collecting private information and leads to less firm-specific information being reflected in stock prices. The cross country study by Morck, Yeung and Yu (2000) relates aspects of the rule of law, such as an efficient judiciary, to stock price synchronicity.

Our second hypothesis is: **H2:** *Stock price synchronicity is lower for firms in provinces with higher quality of law enforcement.*

China's leaders have long recognized the need for political reform, albeit very gradual political reform. While China is still a largely authoritarian communist state, there have been significant movements towards political pluralism (White, 1993, Chapter 8). For example, non-party members such as professional experts, entrepreneurs, and members of minority political parties occupy approximately one third of the seats in the National People's Congress (NPC), the highest legislative body in China. This significantly contributes to elements of pluralism in Chinese political structures and decision making. The delegates to the NPC are elected by the provincial People's Congresses, and the degree of pluralism varies across provinces. Thus we will be able to examine the effect on stock price synchronicity of a more open and pluralistic political structure.

The cross country literature has shown that more democratic political environments are associated with economic growth (Rodrik and Wacziarg, 2006; Borner, Brunetti and Weder, 1995). Shleifer and Vishny (1993) and Mauro (1995) document that poor quality political institutions are associated with substantial economic costs, especially in developing economies. Furthermore, the literature shows a positive association between liberal democracy and successful economic liberalization (e.g., Zack-Williams, 2001).

Increasing numbers of non-party members are elected to both national and provincial People's Congresses. New members include many owners of private firms who have a motivation to promote the recognition and protection of private-property rights. If there is a higher proportion of non-Communist Party members in the provincial People's Congress than in the national People's Congress, we suggest that the province has a more pluralistic political environment. A more pluralistic political environment could lead to improved decision making due to increased debate and monitoring. This mechanism is similar to the competition among political parties in western countries; it improves the quality of the local government by reducing the uncertainty and opacity of political decision making. Thus, greater political pluralism in a province will reduce the costs resulting from political risk to collect private information which decreases synchronicity. Morck, Yeung, and Yu (2000) linked one aspect of the country-level quality of government, an absence of corruption, to stock price synchronicity.

Our third hypothesis is: **H3:** *Stock price synchronicity is lower for firms in provinces with more political pluralism.*

3 Data

3.1 Sample

Our sample covers the period 1998-2007 and includes all non-financial firms in China with available stock returns data from DataStream for at least 200 trading days in a particular year.² Altogether, the sample includes 1012 firms and 5570 firm-year observations.³ Firms are assigned to the province where the company is incorporated. Each firm's location was collected by hand from the prospectus. Importantly, there is no evidence that firms are concentrated in particular regions; there are as many provinces (five) with 15-20 firms as there are with 60-65 firms.⁴

3.2 Measurement of stock price synchronicity

Stock price synchronicity is based on annual market model regressions for each firm where the daily stock returns are regressed on the returns of an industry index and the market index. That is, we estimate the following market model for each stock and each year using daily returns data:

$$Return_{it} = \alpha + \beta_1 Industry Return_t + \beta_2 Market Return_t + \varepsilon_{it}$$
(1)

where $Return_{it}$ denotes the daily return for firm *i* and day *t*. IndustryReturnt is the industry return calculated as a value-weighted stock return of all other firms within the same industry as firm *i* (with *Return_{it}* omitted); *MarketReturn_t* is the value-weighted return for all firms; and ε_{it} represents unspecified random factors.

The Chinese market has a unique feature: there are shares issued to domestic investors (A-shares) and two types of shares issued to foreign investors (B-shares that trade on the Shanghai or Shenzhen stock exchanges and H-shares that are traded in Hong Kong). Equation (1), as specified above, uses A share returns. We also estimate market models that include the influence of foreign investors through B or H share trading. We follow the approach in Gul, Kim, and Qiu (2010) to account for differences among the

 $^{^{2}}$ We begin in 1998 because the province level data on the institutional environment are not available earlier and we end before the onset of the global financial crisis.

³ We exclude the observation for the year in which a firm concludes its IPO.

⁴ There are 22 provinces, 5 autonomous regions, and 4 municipalities. The Special Administrative Regions are not included in the sample.

various markets where Chinese shares are traded. For firms with only domestic A-shares, the extended model is equation (2a) which adds the world market returns:

$$Return_{it} = \alpha + \beta_1 IndustryReturn_t + \beta_2 MarketReturn_t + \beta_3 WorldMarketReturn_t + \varepsilon_{it}$$
(2a)

*WorldMarketReturn*_t is the world market return calculated using the MSCI World index for day *t*. For firms with A and B-shares, we also add the value-weighted B-share market return *BShareMarketReturn*_t as in equation (2b):

$$Return_{it} = \alpha + \beta_1 IndustryReturn_t + \beta_2 MarketReturn_t + \beta_3 BShareMarketReturn_t + \beta_4 WorldMarketReturn_t + \varepsilon_{it}$$
(2b)

And for firms with A and H-shares, we add the world market factor and the H-share market returns as in equation (2c):

$$Return_{it} = \alpha + \beta_1 IndustryReturn_t + \beta_2 MarketReturn_t + \beta_3 HShareMarketReturn_t + \beta_4 WorldMarketReturn_t + \varepsilon_{it}$$
(2c)

where $HShareMarketReturn_t$ is the value-weighted Hong Kong market return. Thus, the extended market model is estimated for each firm for each year using (2a) or (2b) or (2c).

The R_i^2 of the market model estimated for a particular firm in a particular year is our measure of its stock price synchronicity. A higher value of R_i^2 means higher stock price synchronicity with market movements and less firm-specific return variation.⁵ Because R_i^2 is bounded within [0, 1], ψ_i , the logistic transformation of R_i^2 , is our preferred measure of the annual stock price synchronicity for firm *i*:

$$\psi_i = \log(\frac{R_i^2}{1 - R_i^2})$$

We will denote synchronicity from the domestic share model, equation (1), as $R_i^2(1)$ and synchronicity from the extended model, equations (2a-b-c), by $R_i^2(2)$. The corresponding logistic transforms are $\psi(1)$ and $\psi(2)$.

⁵ A higher value of R_i^2 is also associated with larger estimates of β from the market model, which is often interpreted as risk. Thus, increased synchronicity is the same as increased risk.

3.3 Measures of legal and political institutions

We develop three measures of province-level characteristics in China, each associated with one of our hypotheses: property rights protection, rule of law, and political pluralism. A major challenge in our study is that direct measures of these phenomena are not available; only imperfect proxies exist. Following on the work of Hasan, Wachtel, and Zhou (2009), we were able to collect proxy measures from a variety of Chinese sources.

The variable *PropertyRights* is the number of domestic trademark applications per firm for a certain province and year. Firms will make use of trademarking if they are confident that the institutional environment in the province is good enough to actually protect the property rights that come with a trademark. Thus, a higher value of *PropertyRights* represents a higher level of property rights protection. The data on domestic trademark applications is from the annual issues of the *Almanac of China's Property Rights* and the *Yearbook of China's Industrial and Commercial Administrative*, annual provincial yearbooks, and the government-sponsored trademark website, *China Trademark Online*. When the data are missing in a certain year, we use the product between the national data in that year and the proportion of applications from the province in 1998.

The variable *RuleofLaw* is the number of lawyers per 10,000 people for a certain province and year. An increased presence of legal professionals in a province is associated with both the development of legal institutions and of the mechanisms for law enforcement. Thus, a higher value of the variable *RuleofLaw* represents a higher level of law enforcement. The number of lawyers in each province is taken from the *Statistics Yearbook of China's Legislation* and the annual issues of the *Statistics Yearbooks* of each province.⁶ It is supplemented by additional information from Chinese web based resources such as the China Lawyering (www.china-lawyering.com) and China Lawyers Investigation (www.007cn.cn). If the data are missing, we use interpolated values based on nationwide growth in the number of lawyers. Population data are from the National Bureau of Statistics of China.

The variable *PoliticalPluralism* is the proportion of non-Communist party members in the provincial People's Congress relative to the proportion in the National People's Congress. If the provincial proportion of non-Communist Party members is higher than the national benchmark at that time, then the province arguably has a more open or pluralistic

⁶ They provide data for 1990, 1995, and 2000–2002.

political environment. Thus, a higher value of the variable *PoliticalPluralism* indicates a higher level of political pluralism. The data on political pluralism are taken from the regional *People's Congress Yearbooks* of each province in China and the Examination and Approval Reporting Document issued by the Examination Committee of the People's Congress. Information on the membership structures of the People's Congress in six provinces is not available, and so data from neighboring provinces with similar political characteristics are used as an estimate.

3.4 Other variables

The set of control variables includes characteristics of firms and the macro environment that are known to influence synchronicity (Chan and Hameed, 2006; Gul, Kim and Qiu, 2010). Specifically, we include the proportion of equity held by the government and by foreign owners (GovernmentOwn and ForeignOwn) for each firm-year. The government and foreign ownership data are from the NUS Business School's database of Chinese listed firms' ownership structures. We also include dummy variables for the share structure (BShare and HShare if the firm has issued B shares and H shares respectively).⁷ Firm characteristics included are: a) AccountingOpacity, the ratio of the absolute value of accounting accrual to total assets, where accrual is equal to earnings after extraordinary items plus depreciation minus cash flow from operations; b) the annual trading turnover (Volume), which is the total number of shares traded in a year divided by the total number of shares outstanding at the end of the fiscal year; c) Size, the logarithm of total assets; (d) Leverage, total liabilities divided by total assets; (e) EarningVolatility, the standard deviation of a firm's ROA for the preceding five-year period, including the current year; (f) MatketToBook, the ratio of market value of assets to the book value of assets, where the market value of assets is defined as the book value of assets minus the book value of equity plus the market value of equity; and (g) Log(IPOAge), the logarithm of the number years from the IPO. The stock turnover data are from DataStream and the firm accounting data are from the Worldscope database

⁷ Firms with B-shares and H-shares must report in compliance with International Financial Reporting Standards or Hong Kong Generally Accepted Accounting Principles. The stricter disclosure rules would improve firms' information environments and reduce stock price synchronicity.

We also include two measures of industry size: (a) the logarithm of the number of firms in the industry to which a firm belongs (*IndustryByNumber*) and (b) the logarithm of the total assets of all firms in the industry that are in our sample (*IndustryBySize*). Industries are defined according to the one digit industry code from the Worldscope database. Macro variables in the set of controls are (a) the annual province-level growth rate in per capita real GDP (*GDPGrowth*), and (b) the per capita real GDP (*GDPPerCapita*). The real annual per capita GDP for each province is from China Economic Information Network Database. Table 1 presents a summary of variable names and definitions.

3.5 Descriptive statistics

The mean value of the three institutional variables and the mean value of stock price synchronicity measured by $R^2(1)$ by province are shown in Figure 1.⁸ Most of the provinces with higher institutional development have low stock price synchronicity. For example, Gan'Su province has the highest stock price synchronicity. Moreover, it has the lowest value for *PropertyRights* and the second lowest value for *RuleofLaw* and *PoliticalPluralism*. Bei'Jing has the lowest stock price synchronicity; it has the highest value of *PropertyRights* and *RuleofLaw* and the fourth highest value of *PoliticalPluralism*.

Summary statistics for the whole sample are shown in Table 2. All variables are winsorized at the 1 and 99 percentiles to exclude possible outliers. As shown in the table, the mean value of $R^2(1)$ is 0.490 while the mean value of $R^2(2)$ is 0.432.⁹ The large standard deviations and wide ranges of both R^2 and ψ indicate that there is considerable crosssectional variation in synchronicity. All of the institutional variables exhibit a reasonable amount of variation across time and province. On average, the state shareholder holds 31.0 percent and foreign shareholders hold 6.7 percent of shares outstanding, suggesting that the government still has a dominant impact on Chinese listed firms.

Table 3 presents the correlations between our province-level and firm-level variables. The correlation between the two synchronicity measures, $\Psi(1)$ and $\Psi(2)$, is 0.975. Measures of synchronicity are negatively correlated with property rights (*PropertyRights*), rule of law (*RuleofLaw*), and political pluralism (*PoliticalPluralism*). A firm's stock price

⁸ The values are the means for all the available observations of companies in the province.

⁹ These statistics are similar to the results reported with cross country data by Morck, Yeung and Yu (2000), as well as in the study of China by Gul, Kim and Qiu (2010).

synchronicity is lower when it is located in a province in which there is sounder institutional development. The simple correlations are consistent with our three hypotheses.

4 Baseline regression results

To test for the effects on synchronicity of property rights (H1), of rule of law (H2), and of political pluralism (H3), we estimate the following equation:

 $\psi_{it} = \alpha + \beta_1 PopertyRights_{it} + \beta_2 RuleofLaw_{it} + \beta_3 PoliticalPluralism_{it} + (CONTROL_{it}) + (YearDummies) + (IndustryDummies) + \varepsilon_{it}$ (3)

where ψ_{it} is the measure of stock synchronicity for firm *i* and year *t*. The institutional environment variables *PropertyRights*, *RuleofLaw*, and *PoliticalPluralism* are our primary interests and we expect that β_1 , β_2 and $\beta_3 < 0$. CONTROL denotes a set of control variables, including the variables *GovernmentOwn*, *ForeignOwn*, *BShare*, *HShare*, *AccountingOpacity*, *Volume*, *Size*, *Leverage*, *EarningVolatility*, *MarketToBook*, *Log(IPOAge)*, *IndustryByNumber*, *IndustryBySize*, *GDPGrowth*, and *Log(GDPPerCapita)*. Finally, the regressions include fixed effects for the year (*YearDummies*) and industry (*IndustryDummies*).

Government related shareholders are more likely to be associated with expropriation risk and information asymmetry problems compared to foreign shareholders. Thus, we expect the coefficient on *GovernmentOwn* to be positive and the coefficient on *ForeignOwn* to be negative. Since foreign investors are presumably more skilled at collecting and trading on firm-specific information, we expect that the variables *BShare* and *HShare* are negatively correlated with stock price synchronicity. We expect that the coefficient on *AccountingQpacity* is negative because firms with higher accounting opacity have a worse information environment. We expect that the coefficient on *Volume* is negative because more trading will help incorporate more firm-specific information. We expect that the coefficient on the variable *Size* is positive because stocks of large firms are more likely to be aligned with the whole market. The coefficient on the variable *Leverage* is likely to be positive because the cost of collecting private information may be higher for firms with greater risk of financial distress. We expect that the coefficient on *EarningVolatility* is negative because stocks of firms with higher earning uncertainty have more firm-specific variation. The coefficient on the variable *MarketToBook* is expected to be negative because stocks of firms with higher growth potential are likely to incorporate more firm-specific information. We expect that the coefficient on Log(IPOAge) is positive because more of the time-invariant firm-specific information such as managerial quality is already incorporated in the stock price of older firms and there is generally less firm-specific information that is not yet reflected in the stock price (Dasgupta, Gan, and Gao, 2010).

Estimates of Equation (3) with $\psi(1)$ as the dependent variable are shown in Table 4. The t-statistics are calculated using robust standard errors corrected for firm-level clustering which may result from serial dependency in the firm data. The first three columns in the table show each of our key institutional variables separately estimated with OLS. The last two columns include all of the variables first estimated with OLS and then with GMM. Since simultaneity among the variables is a potentially serious problem, equation (5) of Table 4 shows estimates of the full equation using the two-step system GMM estimator. We treat all institutional variables and control variables excluding IndustryByNumber, IndustryBySize, GDPGrowth, and Log(GDPPerCapita) as endogenous and the other variables as exogenous. We also add another instrument, RelativeDistance, which is defined as the distance from Beijing to each of the provincial capitals scaled by the longest distance from Beijing to a provincial capital. This variable is chosen because firms are less likely to be influenced by the central government if they are located in a province that is far from Beijing. We use a Hansen test for the overall validity of our instruments; the null hypothesis is that the instruments as a group are exogenous. The p-value of the Hansen test is larger than 0.1, suggesting our instruments are exogenous.¹⁰ GMM estimates are not shown for the other specifications because the estimates are over identified.

In all the results shown, we find that stock price synchronicity is lower when there is sounder institutional development in term of property rights protection, rule of law, and political pluralism. That is, our measures of institutional development all have the expected negative impact on synchronicity and, moreover, the coefficients are all statistically significant.

The effects of these institutional factors on stock price synchronicity are also economically relevant. We use the results reported in Column (4) of Table 4 to assess the impact of the institutional variables on stock price synchronicity. A one standard deviation

¹⁰ The p- value for the Hansen test for equation (5) is 0.191 and the values for the AR(1) and AR(2) tests are 0.000 and 0.925, respectively.

increase in *PropertyRights* decreases stock price synchronicity (i.e. $\psi(1)$) by 0.154*0.484 or 0.075, which is about 2.5 percent of the range of synchronicity (the difference between the largest and smallest values of $\psi(1)$ is 2.993). A one standard deviation increase in *RuleofLaw* decreases stock price synchronicity by 0.050*0.715 or 0.036, roughly 1.2 percent of the range. Finally, a one standard deviation increase in *PoliticalPluralism* decreases stock price synchronicity by 0.119*0.119 or 0.014, just over 0.5 percent of the range. Taken together, these examples underline the importance that different institutional factors have for stock price synchronicity.

All the control variables have the expected signs in the OLS estimates, though there are some exceptions with the GMM estimator. The coefficient on *GovernmentOwn* is significantly positive, suggesting that higher state ownership is associated with less use of firm-specific information in determining stock prices. The coefficient on the variable *ForeignOwn* is significantly negative, suggesting that higher foreign ownership increases the use of firm-specific information. However, this result is not found with the GMM estimator. The coefficients of the variables *Volume* and *MartketToBook* are significantly negative, but only with OLS. Firms with high growth opportunity tend to commove less. The coefficient estimates of the remaining variable generally have the expected signs, but are insignificant.

The significant negative signs on *BShare* in Table 4 indicate that the presence of foreign investors increases the information content of stock prices. The degree of synchronicity of these firms is likely to differ from those restricted to the domestic stock market for at least two reasons. First, compared with firms that exclusively issued A-shares, these firms have stricter financial reporting regulation and their financial statements are more likely to be audited by international auditing firms. Second, foreign investors are more likely to have superior capabilities, resources, and skills to collect and process firm-specific information (Gul, Kim and Qiu, 2010).

Since the number of firms with either B-shares (available to foreigners on the Shanghai or Shenzhen stock exchanges) or H-shares (trading in Hong Kong) is relatively small, we do not estimate the relationships separately. Instead, we use the measure of synchronicity from the extended market model (2a-b-c) to account for international market influences. Table 5 presents a summary of regression results that use synchronicity calculated from the extended market model, $\psi(2)$, as the dependent variable. Each equation

includes all the control variables shown in Table 4; to conserve space, only key coefficients are shown in the table. The coefficient estimates in Table 5 are qualitatively similar to those reported in Table 4. The presence of foreign traders is important but does not obviate the significance of local institutions.

5 Model extensions and robustness checks.

5.1 Alternative measures of institutional quality

Our proxies for institutional quality reflect our efforts to measure the relevant characteristics directly. In this section, we compare our constructed variables to published indices that measure related aspects of the institutional development of Chinese provinces. We were able to locate two such measures; although the data are limited, they are sufficient to test the robustness of our results.

The first measure is an index of administrative transparency constructed since 2009 by the Beijing University Center for Public Participation Studies and Supports (China Law Center, 2011). It is an assessment of local governments with regard to the transparency of government communication and efforts to build government systems. For the lack of any earlier data, we use the 2009 administrative transparency index for all years. The second measure is a broader index of marketization constructed by the National Economic Research Institute (NERI, 2010). The marketization index is available annually from 1997-2007. The index is based on a large number of indicators that reflect institutions and the economic structure. Among the elements of the index are the extent of state ownership, barriers to inter-regional trade, price controls, the environment for FDI and the financial sector, labor mobility, and the legal framework (including a measure of patent registration).

These indexes are far from perfect for at least two reasons. First, they weight diverse bits of information with an arbitrary scaling; it is unclear how to properly interpret a unit increase in the index. Second, the actual components of the indices are not provided in the sources, so their meaning is opaque. Nevertheless, we add the indexes to our baseline regressions as a robustness test.

Table 6 provides a summary of results; only coefficients for the institutional variables are shown. Each equation includes all the control variables shown in Table 4. The

19

scaled marketization index is the province's marketization score relative to the average for all provinces in that year. We show each index separately and with our three proxy measures of institutional quality. There is a strong negative relationship between synchronicity and the marketization and the administrative transparency indexes. Furthermore, the coefficients are reduced somewhat when we also include our proxy measures for property rights, rule of law, and pluralism. Importantly, there is little difference in the coefficients on our three institutional proxies between the baseline estimates (column (4) Table 4) and the estimates in Table 6 that include the overall institutional indexes. After controlling for these additional institutional variables, our main results still hold.

A final robustness check is based on the changing nature of our political pluralism variable. In 2002, the Party revised its Charter to allow private entrepreneurs to join. Since then, many such "red capitalists" became members of the Chinese Community party and even members in the People's Congress, which has contributed to pluralism (Dickson, 2003; Chen and Dickson, 2008). In a sense, this reform represents an exogenous shock which might influence stock price synchronicity. In results not shown here, we add a dummy variable for this post reform era to our baseline equation. The coefficient on the dummy is significantly negative, suggesting that stock price synchronicity decreased after the party charter revision that allowed private entrepreneurs to join; furthermore, the coefficients on the institutional variables are largely unaffected.

5.2 Threshold effects

Morck, Yeung, and Yu (2000) find that the relationship between investor protection and stock price synchronicity is different in emerging markets as compared to developed countries. To see if there are any similar threshold effects within China, we divide our sample into provinces with above and below median GDP per capita. We estimate synchronicity regressions for each sub sample and the results are summarized in Table 7. The coefficient estimates across these two sub samples are qualitatively identical to those reported in Table 4, which thus support our previous conclusions and indicate that our results are not driven by the differences between more and less developed provinces.

5.3 Alternative explanations

Throughout this paper, we have maintained that poor institutional development increases the cost of information collection and reduce investors' incentives to collect private information. This reduces the information content of stock prices and increases stock price synchronicity. However, there may be other explanations of our main results. For example, Khanna and Thomas (2009) use a unique dataset in Chile and show that two firms with more connections are more likely to have synchronized stock returns. Poor provincial institutions may enhance connectedness among managers, resulting in the development of business groups, vertical integration, and operating diversification, all of which further increase stock price synchronicity. Another possibility is that regions with stronger institutions have a bigger investor pool. If investors favor firms in their home province then stock price synchronicity is lower.

In the only other study of synchronicity in China, Gul, Kim and Qiu (2010) argue that state ownership is associated with inefficient corporate governance and less protection for minority shareholders. Better investor protection increases investors' incentives to collect private information, which will result in less synchronicity. Thus, we expect that the effect of institutional development on synchronicity is more pronounced for firms with higher government ownership.

To formally address this issue, we add interaction terms between our institutional variables and the state ownership variable in the synchronicity regressions. The equations are estimated with OLS and include all the standard control variables shown in Table 4. The results are summarized in Table 8, which shows that the interactions between government ownership and the variables *PropertyRights* and *RuleofLaw* are negative and significant. This is true when all three variables and the interaction terms are in the regression together (column (4) of Table 8). That is, the (negative) effect of institutional quality on synchronicity is more pronounced for firms with higher government ownership.

Next we test whether provincial institutional development has a more significant impact on synchronicity for firms with lower foreign ownership. To do so, we add interaction terms between our institutional development variables and the foreign ownership variable to the synchronicity regressions. Foreign shareholders are less likely to be associated with expropriation risk and information asymmetry problems (Kang and Stulz, 1997; Jiang and Kim, 2004). If, in fact, the higher investor protection provided by the better institu-

tional environment decreases the cost of information collection and increases investors' incentives to collect private information, which contributes to the incorporation of information into equity prices, we expect to find that the effect of institutional development on synchronicity is more pronounced for firms with lower foreign ownership and that there is a positive coefficient on this interaction variable. The estimates in columns (5) to (8) of Table 8 show that the coefficients on the interactions between foreign ownership and the three institutional development variables are all positive and significant. This is true when the interactions with the three institutional variables are considered individually or when all three variables and the interaction terms are entered into the regression together in column (8) of Table 8. That is, increased foreign ownership decreases the size of the negative effect of institutional quality on synchronicity.

In summary, we find that better institutions have a more pronounced effect on stock price informativeness for firms with higher government ownership and lower foreign ownership.¹¹ These results suggest that the poor institutional development provides less investor protection. This increases the cost of information collection and reduces investors' incentives to collect private information. This further reduces the information content of stock prices and increases stock price synchronicity.

In addition, West (1988) and Siew, Yang, and Zhang (2010) suggest that low stock price synchronicity may be associated with a poor information environment, which questions the validity of stock price synchronicity to measure stock return informativeness. Veldkamp (2006) provides a theoretical model to show that low stock price synchronicity is a good measure of a better information environment and that the empirical studies cited are supportive (e.g., Morck, Yeung, and Yu, 2000). If low stock price synchronicity means that there is more information in stock prices, then investment should be more sensitive to lagged company valuation for firms with lower stock price synchronicity.

We can explore this specifically with our data by estimating equation (4):

 $Investment = \alpha + \beta_1 Q + \beta_2 CashFlow + \beta_3 Q * DR\Psi(1) + \beta_4 Q * DRAccountingOpacity + (YearDummies) + (IndustryDummies) + \varepsilon_{it}$ (4)

where the dependent variable, *Investment*, is capital expenditure scaled by beginning-ofyear total assets. *Q* is equal to *MarketToBook* at the end of the previous year. Although

¹¹ The interpretation of the interaction terms is symmetric; that is, higher foreign ownership and lower state ownership matter more for stock price synchronicity in provinces with poor institutions.

this approach follows a q-theory of investment, we also hold cash flow constant to account for financing constraints. *CashFlow* is the sum of income before extraordinary terms and depreciation, net of cash dividends, scaled by beginning-of-year total assets. $DR\Psi(1)$ is the scaled decile rank score of $\Psi(1)$. *DRAccountingOpacity* is the scaled decile rank score of *AccountingOpacity*. To obtain these scaled decile rank scores, we classify the observations in each year into deciles from 0 to 9 and then scale these ranks to range between 0 and 1. Table 9 presents estimates of equation (4). The *Q* and *CashFlow* coefficients are significantly positive, reflecting that the company valuation and free cash flow are important determinants of investment. Moreover, the coefficients on $Q*DR\Psi(1)$ are significantly negative, which implies that investment is more sensitive to lagged company valuation for firms with lower stock price synchronicity. The results suggest that managers depend on the information content of stock prices about the firms' fundamentals to make their investment decisions, and so lower stock price synchronicity is a good measure of stock return informativeness. These findings are consistent with the results reported in Chen, Goldstein and Jiang (2007).

5.4 Additional robustness checks

In this section we summarize some additional robustness tests that are not shown here for the sake of space, though they are available from the authors on request.

First, estimates of the market model are based on daily returns and a potential bias can be introduced by non-synchronous trading because some stocks do not trade every day (Scholes and Williams, 1977; French, Schwert, and Stambaugh, 1987). In order to account for this, we also estimated the market models with lagged industry and market returns without any noticeable effect on the results. In addition, we estimated the market models with weekly rather than daily data, which does not alter our results and conclusions.

Second, we varied the windows used to winsorize the data to 2 and 98 percentiles or 5 and 95 percentiles. Our results are still robust to such variation in the size of the data set.

Third, some variables, such as *Ruleoflaw*, have interpolated values for certain years. To ensure that interpolation of this variable does not affect our results, we use the firm-year observations without those interpolated values to estimate our base line synchronicity regression. We obtain similar results and the same conclusions.

Fourth, four large municipalities (Beijing, Chongqing, Shanghai, and Tianjin) are under much stronger control by the central government than the other provinces. Thus, we re-estimated the models without these four municipalities, and find that our results are robust. Similarly, we find that our results are robust when the institutional variables are clustered at the region level.

6 Conclusions

China has experienced dramatic changes over the last 20 years in the establishment of secure property rights, the enhancement of law enforcement, and the liberalization of political institutions with varied experience across China's 31 provinces. China provides a unique opportunity to examine the effects of institutions within one national environment, rather than the more common cross country panel approach. Specifically, this paper investigates whether and how stock price synchronicity is associated with province-level institutional characteristics unique to China.

The province-level institutional variables we examine are two measures of legal development—property rights and the rule of law—and political pluralism. Our evidence indicates that improved property rights, better law enforcement, and greater political pluralism are all associated with more stock price informativeness. Additionally, better institutions have a more pronounced effect on stock price informativeness for firms with higher government ownership and lower foreign ownership.

Our evidence has some clear policy implications for emerging economies which suffer from high stock price synchronicity. We know from Wurgler (2000) and Durnev, Morck, and Yeung (2004), among others, that the efficiency of capital allocation and investment will be improved if stock prices incorporate more firm-specific information (i.e., synchronicity is reduced). And our results show convincingly that institutional improvements and political openness are associated with reduced stock price synchronicity, thus establishing a clear channel by which improvements in the institutional environment are associated with improved economic outcomes.

References

- Acemoglu, D., S. Johnson and J. Robinson. 2005.Institutions as a Fundamental Cause of Long-Run Growth, chapter 6 in P. Aghion and S. N. Durlauf, Eds., Handbook of Economic Growth, Elsevier, Volume 1, Part A, Pages 385-472.
- Ball, R. 2001. Infrastructure requirements for an economically efficient system of public financial reporting and disclosures. Brookings Papers on Financial Services: 127-182.
- Borner, S., A. Brunetti, and B. Weder, 1995. Political Credibility and Economic Development. St. Martin's Press, New York.
- Burns, J.P., 1999. The People's Republic of China at 50: National political reform. The China Quarterly 159, 580–594 (Special Issue: The People's Republic of China after 50 Years).
- Chan, K., A. Hameed, 2006. Stock price synchronicity and analyst coverage in emerging markets. Journal of Financial Economics, 80(1): 115-147.
- Chen, J., and Dickson, B., 2008, Allies of the state: democratic support and regime support among China's private entrepreneurs, China Quarterly, 196: 780–804.
- Chen, Q., and I. Goldstein, and W. Jiang, 2007, Price informativeness and investment sensitivity to stock price, Review of Financial Studies, 20 (3), 619-650
- Cull, R. and L.C. Xu, 2005. Institutions, ownership, and finance: The determinants of profit reinvestment among Chinese firms. Journal of Financial Economics 77 (1): 117-146.
- China Law Center, Yale Law School, 2011. Summary of the 2009 Annal Report on China's Administrative Transparency. http://www.law.yale.edu/documents/pdf/Intellectual_Life/ch_OGIWA_2009_ Report_Exec_Summary_English_FINAL__1-28-11_.pdf
- Dasgupta, S., J. Gan, and N. Gao, 2010. Transparency, stock return synchronicity, and the informativeness of stock prices: theory and evidence. Journal of Financial and Quantitative Analysis 45(5), 1189-1220.
- Dickson, B., 2003. Red Capitalists in China: The Party, Private Entrepreneurs, and Prospects for Political Change, New York, Cambridge University Press.
- Djankov, S., Y. Qian, G. Roland, and E. Zhuravskaya, 2006. Who are China's entrepreneurs? American Economic Review 96 (2), 348-352.
- Du, J. L. and C. G. Xu, 2009. Which firms went public in China? A study of financial market regulation. World Development 37(4), 812-24.
- Durnev, A., R. Morck, and B. Yeung, 2004. Value-enhancing capital budgeting and firmspecific stock return variation. Journal of Finance 59(1): 65-105.
- Fan, J. and T.J. Wong, 2005. Do external auditors perform a corporate governance role in emerging markets? Evidence from East Asia. Journal of Accounting Research 43 (1): 35-72.
- Fernandes, N., and M.A. Ferreira, 2009. Insider trading laws and stock price informativeness. Review of Financial Studies 22 (5): 1845-1887.

- French, K., G.W. Schwert, and R. Stambaugh, 1987. Expected stock returns and volatility. Journal of Financial Economics 19: 3-30.
- Jiang, L. and J.B. Kim, 2004. Foreign equity ownership and information asymmetry: evidence from Japan. Journal of International Financial Management and Accounting 15(3): 185-211.
- Glaeser, E. L., R. La Porta, F. Lopez-de-Silanes and A. Shleifer. 2004. Do Institutions Cause Growth? Journal of Economic Growth 9(3): 271-303.
- Gould, D. M., W.C. Gruben, 1996. The role of intellectual property rights in economic growth. Journal of Development Economics 48 (2): 323-350.
- Grossman, S., and J. Stiglitz, 1980. On the impossibility of informationally efficient markets, American Economic Review 25: 393-408.
- Gul, F., J. Kim, and A.A. Qiu, 2010. Ownership concentration, foreign shareholding, audit quality, and stock price synchronicity: evidence from China. Journal of Financial Economics 95(3): 425-442.
- Hasan, I., P. Wachtel, and M. M. Zhou, 2009. Institutional development, financial deepening and economic growth: evidence from China. Journal of Banking and Finance 33: 157-170.
- Khanna, T. and C. Thomas, 2009. Synchronicity and firm interlocks in an emerging market. Journal of Financial Economics 92 (2): 182-204.
- Krug, B., and H. Hendrischke, 2003. China incorporated: Property rights, networks and the emergence of a private business sector in China. Managerial Finance 29 (12): 32-44.
- Kang, J.K. and R. Stulz, 1997. Why is there a home bias? An analysis of foreign portfolio equity ownership in Japan. Journal of Financial Economics 46: 3-28.
- Mauro, P., 1995. Corruption and growth. Quarterly Journal of Economics 110 (3), 681-712.
- Morck, R., B. Yeung, and W. Yu. 2000. The information content of stock markets: why do emerging markets have synchronous stock price movements? Journal of Financial Economics 58(1): 215-260.
- Jin, L., and S. Myers. 2006. R2 around the world: new theory and new tests. Journal of Finance 79: 257-292.
- National Economic Research Institute, 2010. NERI Index of Marketization of China's Provinces, 2009 Report. (in Chinese) Economic Science Press.
- Pagano, M., and P.F. Volpin. 2005. The political economy of corporate governance. American Economic Review 95 (4): 1005-1030.
- Park, W.G., and J.C. Ginarte, 1997. Intellectual property rights and economic growth. Contemporary Economic Policy 15: 51-61.
- Prasad, E.S., and R.G. Rajan, 2006. Modernizing China's growth program. American Economic Review 96 (2): 331-336.
- Rodrik, D., and R. Wacziarg, 2006. Do democratic transition produce bad economic outcome? American Economic Review 95 (2), 50-55.

- Shleifer, A. and R. Vishny, 1993. Corruption. Quarterly Journal of Economics 108 (3): 599-617.
- Scholes, M. and J. Williams. 1977. Estimating betas from nonsynchronous data. Journal of Finance 5: 309-328.
- Teoh, H.S., Y. Yang, and Y. L. Zhang, 2010. R-square and market efficiency. Working Paper, University of California, Irvine and Chinese University of Hong Kong.
- Veldkamp L., 2006. Information markets and the comovement of asset prices. Review of Economic Studies. 73(3): 823-845
- Wachtel, P. 2011. The evolution of the finance growth nexus, Comparative Economic Studies 53 (3): 475-88.
- West, K. 1988. Dividend innovations and stock price volatility. Econometrica 56: 37-61.
- White, G., 1993. Riding the Tiger, the Politics of Economic Reform in Post-Mao China. Stanford University Press, Stanford.
- Williamson, O. E., 1996. The Mechanisms of Governance. Oxford University Press, New York.
- Wurgler, J., 2000. The financial markets and the allocation of capital. Journal of Financial Economics 58: 187-214.
- Zack-Williams, A.B., 2001. No democracy, no development: Reflections on democracy & development in Africa. Review of African Political Economy 27: 213–223.

Tables and figure

Table 1Variable definitions

$R^{2}(1)$	The R^2 of the market model in equation (1).
$R^{2}(2)$	The R^2 of the market models in equation (2).
Ψ(1)	Logarithmic transformation of $R^2(1)$
Ψ(2)	Logarithmic transformation of $R^2(2)$
PropertyRights	The number of trademark applications per firm in a province; proxy for the awareness of property rights.
RuleofLaw	The number of lawyers per 10,000 people in a province; proxy for rule of law.
PoliticalPluralism	The proportion of non-Communist party members in the provincial People's Congress rela- tive to the proportion in the National People's Congress
GovernmentOwn	The percentage of shares held by government owner(s) at year beginning
ForeignOwn	The percentage of shares held by foreign owner(s) (Hong Kong, Taiwan, other countries) at year beginning
Bshare	A dummy variable which equals 1 if a firm issues B-shares and 0 otherwise
Hshare	A dummy variable which equals 1 if a firm issues H-shares and 0 otherwise
AccountingQpacity	The absolute value of accounting accrual, accrual is equal to earnings after extraordinary items plus depreciation minus cash flow from operations, and then divided by total assets
Volume	The total number of shares traded in a year, scaled by the total number of shares outstand- ing at the end of fiscal year
Size	The logarithm of total assets
Leverage	Total liabilities divided by total assets
EarningVolatility	The standard deviation of a firm's ROAs over the preceding five-year period, including the current year
MarketToBook	Market value of assets over book value of assets. Market value of assets are measured as book value of assets minus book value of equity plus market value of equity
Log(IPOAge)	The logarithm of the firm age since IPO
IndustryByNumber	The logarithm of the number of firms in the industry to which a firm belongs
IndustryBySize	The logarithm of year-end total assets of all sample firms in the industry to which a firm belongs
Investment	Capital expenditure scaled by beginning-of-year total assets.
CashFlow	The sum of income before extraordinary terms and depreciation, net of cash dividends, scaled by beginning-of-year the total assets
Q	MarketToBook at the end of the previous year.
$DR\Psi(l)$	The scaled decile rank score of $\Psi(1)$.
DRAccountingOpacity	The scaled decile rank score of AccountingOpacity
GDPGrowth	Annual growth rate in per capita real GDP
GDPPerCapita	The per capita GDP deflated to the base year of 1998
Log(GDPPerCapita)	The logarithm of per capita GDP deflated to the base year of 1998

Variables	Mean	Std dev	Min	Max
$R^{2}(1)$	0.490	0.176	0.133	0.821
$R^{2}(2)$	0.432	0.174	0.132	0.830
$\Psi(1)$	-0.231	0.340	-1.921	1.072
$\Psi(2)$	-0.332	0.333	-1.921	1.042
PropertyRights	0.472	0.484	0.002	5.011
RuleofLaw	0.691	0.715	0.101	7.451
PoliticalPluralism	1.016	0.119	0.651	1.434
GovernmentOwn	0.310	0.132	0.000	0.716
ForeignOwn	0.067	0.046	0.000	0.399
Hshare	0.039	0.193	0.000	1.000
BShare	0.131	0.337	0.000	1.000
AccountingOpacity	0.202	0.078	0.009	0.438
Volume	2.080	0.716	0.211	3.511
Size	20.670	1.035	19.120	22.812
Leverage	0.385	0.219	0.121	0.962
EarningVolatility	0.203	0.139	0.000	0.492
MarketToBook	2.773	1.566	0.726	5.491
IPOAge	11.645	4.026	2.000	18.000
IndustryByNumber	5.232	1.193	3.261	6.693
IndustryBySize	26.776	1.169	24.145	28.410
Investment	0.169	0.144	0.010	0.526
CashFlow	0.160	0.137	-0.085	0.411
GDPGrowth (%)	7.200	4.526	-17.100	21.574
GDPPerCapita (RMB)	1704.574	1005.678	413.000	12613.000

Table 2Descriptive statistics

Variables are defined in Table 1; data are winsorized at the 1 and 99 percentiles.

Table 3Correlation matrices

Variables are defined in Table 1. The superscripts, ***, **, and * denote the 1%, 5%, and 10% levels of significance, respectively.

Panel A: Province-level variables

	Ψ(1)	PropertyRights	RuleofLaw	PoliticalPluralism	GDPGrowth	Log(GDPPerCapita)
Ψ(1) (average for year) PropertyRights	1.000 -0.524***	1.000				
RuleofLaw	-0.481***	0.489***	1.000			
PoliticalPluralism	-0.358***	0.219***	0.336***	1.000		
GDPGrowth	-0.123**	0.050	0.079	0.160***	1.000	
Log(GDPPerCapita)	-0.110*	0.013	0.079	0.130**	0.727***	1.000

Panel B: Firm-level variables

	Ψ(1)	Ψ(2)	Accounting Opacity	Volume	Size	Leverage	Earning Volatility	Market To Book	IPOAge	Industry By Number	Industry By Size	Investment	Cash Flow
Ψ(1)	1												
Ψ(2)	0.975***	1											
AccountingOpacity	0.288***	0.289***	1										
Volume	-0.312***	-0.313***	-0.928***	1									
Size	-0.011	-0.008	0.001	0.004	1								
Leverage	0.003	0.003	0.016	-0.011	0.009	1							
EarningVolatility	0.003	0.002	-0.007	-0.001	-0.013	0.006	1						
MarketToBook	0.062***	0.063***	0.151***	-0.158***	0.011	-0.008	-0.019	1					
IPOAge	-0.174***	-0.167***	-0.010	0.003	-0.010	-0.007	0.012	0.011	1				
IndustryByNumber	0.004	0.003	0.008	-0.006	0.003	-0.023*	-0.005	-0.013	-0.024	1			
IndustryBySize	0.008	0.006	0.011	-0.012	-0.006	-0.006	-0.015	-0.004	-0.024	0.433***	1		
Investment	-0.045***	-0.039***	0.047***	-0.054***	-0.008	-0.018	0.001	0.189***	0.003	-0.011	-0.005	1	
CashFlow	-0.011	-0.007	0.003	0.002	-0.013	0.004	0.019	0.021	-0.001	-0.005	0.001	0.030**	1

Table 4 The effect of property rights, rule of law, and political pluralism on stock price sychronicity

All variables are defined in Table 1. The dependent variable is $\Psi(1)$. Year and industry dummies are included in all equations and the sample size is 5570. Numbers in parentheses represent absolute values of t-statistics where the standard errors are corrected for clustering at the firm level. The superscripts, ***, **, and * denote the 1%, 5%, and 10% levels of significance, respectively.

	(1) OLS	(2) OLS	(3) OLS	(4) OLS	(5) GMM
PropertyRights	-0.193***			-0.154***	-0.118***
	(16.243)			(12.292)	(7.82)
RuleofLaw		-0.093***		-0.050***	-0.032***
-		(17.091)		(8.940)	(4.04)
PoliticalPluralism			-0.322***	-0.119**	-0.188**
			(7.265)	(2.490)	(2.44)
GovernmentOwn	1.097***	1.175***	1.242***	1.070***	0.734***
	(20.529)	(22.486)	(23.109)	(20.194)	(10.34)
ForeignOwn	-0.541***	-0.534***	-0.483***	-0.563***	0.138
	(5.548)	(5.248)	(4.605)	(5.839)	(1.16)
BShare	-0.159***	-0.178***	-0.194***	-0.154***	-0.265***
	(8.153)	(9.878)	(10.223)	(8.198)	(5.77)
HShare	-0.032	-0.039	-0.043	-0.031	-0.368***
	(1.398)	(1.563)	(1.625)	(1.426)	(3.05)
AccountingOpacity	0.153	0.113	0.099	0.156	0.175
	(1.429)	(1.031)	(0.878)	(1.485)	(0.64)
Volume	-0.030**	-0.036***	-0.040***	-0.028**	-0.040
	(2.494)	(3.045)	(3.291)	(2.456)	(1.44)
Size	-0.002	-0.003	-0.001	-0.003	0.005
	(0.762)	(0.987)	(0.243)	(0.952)	(0.73)
Leverage	0.029**	0.033**	0.036**	0.032**	0.072**
	(2.037)	(2.250)	(2.427)	(2.257)	(2.25)
EarningVolatility	0.001	-0.003	0.007	0.001	0.108**
	(0.024)	(0.153)	(0.321)	(0.049)	(2.30)
MarketToBook	-0.010***	-0.011***	-0.011***	-0.010***	-0.007
	(4.977)	(5.438)	(5.382)	(5.022)	(1.50)
Log(IPOAge)	0.012	0.014	0.019	0.011	-0.008
	(0.822)	(0.942)	(1.186)	(0.796)	(-0.28)
IndustryByNumber	0.004	0.010	0.009	0.006	-0.005
	(0.471)	(1.153)	(1.072)	(0.732)	(0.28)
IndustryBySize	-0.003	-0.005	-0.004	-0.004	0.004
	(0.471)	(0.756)	(0.549)	(0.637)	(0.36)
GDPGrowth	-0.001	-0.004***	-0.003*	-0.001	-0.000
	(0.904)	(2.589)	(1.960)	(0.907)	(0.02)
Log(GDPPerCapita)	-0.052*	-0.010	-0.027	-0.043	-0.062**
	(1.758)	(0.343)	(0.867)	(1.494)	(2.29)
Adj. R^2 or Wald \div^2 for (5)	0.567	0.544	0.518	0.577	616.78

Table 5The effect of property rights, rule of law, and political pluralism on
stock price synchronicity with extended model

All variables are defined in Table 1. The dependent variable is $\Psi(2)$. Each equation contains all the control variables shown in Table 4 as well as year and industry dummies. The sample size is 5570. Numbers in parentheses are the absolute values of t-statistics where the standard errors are corrected for clustering at the firm level. The superscripts, ***, **, and * denote the 1%, 5%, and 10% levels of significance, respectively.

	(1) OLS	(2) OLS	(3) OLS	(4) OLS	(5) GMM
PropertyRights	-0.167***			-0.131***	-0.095***
	(14.058)			(10.955)	(6.17)
RuleofLaw		-0.081***		-0.044***	-0027***
		(14.720)		(7.807)	(3.53)
PoliticalPluralism			-0.301***	-0.127***	-0.185
			(6.777)	(2.684)	(2.42)**
GovernmentOwn	1.082***	1.148***	1.205***	1.056***	0.771***
	(20.567)	(22.364)	(22.930)	(20.228)	(10.76)
ForeignOwn	-0.546***	-0.541***	-0.497***	-0.566***	0.144
	(5.690)	(5.445)	(4.873)	(5.954)	(1.15)
BShare	-0.158***	-0.174***	-0.188***	-0.154***	-0.264***
	(8.808)	(10.496)	(10.799)	(8.878)	(5.99)
HShare	-0.029	-0.035	-0.039	-0.028	-0.346***
Adj. R^2 or	(1.272)	(1.435)	(1.494)	(1.289)	(3.04)
Wald \div^2 for (5)	0.547	0.530	0.510	0.555	625.49

Table 6 Effect of alternative institutional indexes on stock price synchronicity

All variables are defined in Table 1. The dependent variable is $\Psi(1)$. Each equation contains all the control variables shown in Table 4 as well as year and industry dummies. Numbers in parentheses represent absolute values of t-statistics that are adjusted using standard errors corrected for clustering at the firm level. The superscripts, ***, **, and * denote the 1%, 5%, and 10% levels of significance, respectively.

	Administrative transparency		Market	tization	Scaled marketization		
Index	-0.004***	-0.001**	-0.040***	-0.025***	-0.214***	-0.135***	
	(6.819)	(1.969)	(11.589)	(7.527)	(11.322)	(7.565)	
PropertyRights		-0.155***		-0.131***		-0.131***	
		(12.246)		(11.162)		(11.208)	
RuleofLaw		-0.039***		-0.042***		-0.044***	
		(6.375)		(7.687)		(8.064)	
PoliticalPluralism		-0.122***		-0.087*		-0.095**	
		(2.631)		(1.811)		(2.010)	
Sample size	5380	5380	5532	5532	5532	5532	
Adj. R ²	0.527	0.580	0.547	0.589	0.545	0.589	

Table 7Stock price synchronicity in sub samples

Variables are defined in Table 1. The dependent variable is $\Psi(1)$. Each equation includes year and industry dummies as well as all the control variables shown in Table 4. Numbers in parentheses represent absolute values of t-statistics that are adjusted using standard errors corrected for clustering at the firm level. The superscripts, ***, **, and * denote the 1%, 5%, and 10% levels of significance, respectively.

		More develo	oped province	es		Less develo	pped province	s
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
PropertyRights	-0.186***			-0.152***	-0.205***			-0.182***
	(13.876)			(9.914)	(11.346)			(10.643)
RuleofLaw		-0.084***		-0.023***		-0.106***		-0.086***
		(14.319)		(3.057)		(12.630)		(12.315)
PoliticalPluralism			-0.423***	-0.138**			-0.180***	-0.120*
			(7.696)	(2.361)			(2.696)	(1.955)
Sample size	2861	2861	2861	2861	2709	2709	2709	2709
Adj. R^2	0.600	0.570	0.552	0.603	0.533	0.517	0.486	0.555

Table 8The effect of state and foreign ownership on the relation between
institutional development and stock price synchronicity

All variables are defined in Table 1. The dependent variable is $\Psi(1)$. Each equation includes year and industry dummies as well as all the control variables shown in Table 4; only the coefficients on institutional and ownership variables and the interactions are shown in the table. Numbers in parentheses represent absolute values of t-statistics that are adjusted using standard errors corrected for clustering at the firm level. The superscripts, ***, **, and * denote the 1%, 5%, and 10% levels of significance, respectively.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
PropertyRights	-0.051			-0.010	0.312***			0.252***
	(1.605)			(0.309)	(12.452)			(9.646)
RuleofLaw		-0.060***		0.021		- 0.150***		- 0.088***
		(2.867)		(1.005)		(10.223)		(5.802)
PoliticalPluralism			-0.451***	-0.218**			- 0.636***	- 0.357***
			(4.127)	(1.968)			(7.690)	(4.572)
PropertyRights* GovernmentOwn	-0.669***			-0.695***				
	(4.686)			(5.045)				
RuleofLaw * GovernmentOwn		-0.154		-0.329***				
		(1.606)		(3.464)				
PoliticalPluralism * Governmen- tOwn			0 491	0 371				
10 mil			(1.343)	(1.054)				
PropertyRights* ForeignOwn			(1.515)	(1.051)	1 512***			1 282***
					(6 254)			(5.116)
RuleofLaw *					(0.254)			(5.110)
ForeignOwn						0.731***		0.481***
						(4.552)		(2.931)
PoliticalPluralism * ForeignOwn							4.284***	3.342***
							(4.819)	(4.378)
GovernmentOwn	1.358***	1.264***	0.746**	1.151***	1.077***	1.169***	1.244***	1.052***
	(16.252)	(15.434)	(1.988)	(3.246)	(20.264)	(22.483)	(23.206)	(20.068)
ForeignOwn	-0.486***	-0.531***	-0.484***	-0.503***	1.225***	1.005***	- 4.804***	- 4.824***
	1.358***	1.264***	0.746**	1.151***	(8.246)	(7.242)	(5.439)	(6.428)
Sample size	5570	5570	5570	5570	5570	5570	5570	5570
Adj. R ²	0.572	0.545	0.519	0.585	0.572	0.547	0.521	0.584

Table 9The effect of stock price synchronicity on the relation between
investment and stock price

The dependent variable is *Investment*, which is defined as capital expenditure scaled by beginning-of-year total assets. Q is equal to *MarketToBook* at the end of the previous year. $DR\Psi(1)$ is the scaled decile rank score of $\Psi(1)$. *DRAccountingOpacity* is the scaled decile rank score of *AccountingOpacity*. Other variables are defined in Table 1. Each equation also includes year and industry dummies. Numbers in parentheses represent absolute values of t-statistics that are adjusted using standard errors corrected for clustering at the firm level. The superscripts, ***, **, and * denote the 1%, 5%, and 10% levels of significance, respectively.

	(1)	(2)	(3)
Q	0.017***	0.021***	0.021***
	(9.381)	(9.127)	(8.273)
CashFlow	0.027*	0.026*	0.026*
	(1.947)	(1.897)	(1.898)
$Q*DR\Psi(1)$		-0.008***	-0.008***
		(3.494)	(3.438)
Q*DRAccountingOpacity			0.002
			(0.682)
Sample size	5570	5570	5570
Adj. R ²	0.037	0.039	0.039

Figure 1 Institutional development and stock price synchronicity

The figure present the mean values of *PropertyRights*, *Ruleoflaw*, *PoliticalPluralism*, and $R^2(1)$ for each province. All variables are as defined in Table 1.





Rule of Law (The number of lawyers per 10,000 people in a province) - Provincial Distribution





Political Pluralism (The proportion of non-Communist party members in the provincial People's Congress relative to the proportion in the National People's Congress) - Provincial Distribution

R Square - Provincial Distribution



BOFIT Discussion Papers

A series devoted to academic studies by BOFIT economists and guest researchers. The focus is on works relevant for economic policy and economic developments in transition / emerging economies.

- 2012 No 1 Walid Marrouch and Rima Turk-Ariss: Bank pricing under oligopsony-oligopoly: Evidence from 103 developing countries
 - No 2 Ying Fang, Shicheng Huang and Linlin Niu: De facto currency baskets of China and East Asian economies: The rising weights
 - No 3 Zuzana Fungáčová and Petr Jakubík: Bank stress tests as an information device for emerging markets: The case of Russia
 - No 4 Jan Babecky, Luboš Komárek and Zlatuše Komárková: Integration of Chinese and Russian stock markets with World markets: National and sectoral perspectives
 - No 5 Risto Herrala and Yandong Jia: Has the Chinese growth model changed? A view from the credit market
 - No 6 Sanna Kurronen: Financial sector in resource-dependent economies
 - No 7 Laurent Weill and Christophe Godlewski: Why do large firms go for Islamic loans?
 - No 8 Iftekhar Hasan and Ru Xie: A note on foreign bank entry and bank corporate governance in China
 - No 9 Yi Yao, Rong Yang, Zhiyuan Liu and Iftekhar Hasan: Government intervention and institutional trading strategy: Evidence from a transition country
 - No 10 Daniel Berkowitz, Mark Hoekstra and Koen Schoors: Does finance cause growth? Evidence from the origins of banking in Russia
 - No 11 Michael Funke and Michael Paetz: A DSGE-based assessment of nonlinear loan-to-value policies: Evidence from Hong Kong
 - No 12 Irina Andrievskaya: Measuring systemic funding liquidity risk in the Russian banking system
 - No 13 Xi Chen and Michael Funke: The dynamics of catch-up and skill and technology upgrading in China
 - No 14 Yin-Wong Cheung, Menzie D. Chinn and XingWang Qian: Are Chinese trade flows different?
 - No 15 Niko Korte: Predictive power of confidence indicators for the russian economy
 - No 16 Qianying Chen, Michael Funke and Michael Paetz: Market and non-market monetary policy tools in a calibrated DSGE model for Mainland China
 - No 17 Pierre L. Siklos: No coupling, no decoupling, only mutual inter-dependence: Business cycles in emerging vs. mature economies
 - No 18 José R. Sánchez-Fung: Examining the role of monetary aggregates in China
 - No 19 Konstantins Benkovskis and Julia Wörz: Non-price competitiveness of exports from emerging countries
 - No 20 Martin Feldkircher and likka Korhonen: The rise of China and its implications for emerging markets Evidence from a GVAR model
 - No 21 Pierre Pessarossi and Laurent Weill: Does CEO turnover matter in China? Evidence from the stock market
 - No 22 Alexey Ponomarenko: Early warning indicators of asset price boom/bust cycles in emerging markets
 - No 23 Gabor Pula and Daniel Santabárbara: Is China climbing up the quality ladder?
 - No 24 Christoph Fischer: Currency blocs in the 21st century
 - No 25 Duo Qin and Xinhua He: Modelling the impact of aggregate financial shocks external to the Chinese economy
 - No 26 Martin Feldkircher: The determinants of vulnerability to the global financial crisis 2008 to 2009: Credit growth and other sources of risk
 - No 27 Xi Chen and Michael Funke: Real-time warning signs of emerging and collapsing Chinese house price bubbles
 - No 28 Yi David Wang: Convertibility restriction in China's foreign exchange market and its impact on forward pricing
 - No 29 Risto Herrala and Rima Turk Ariss: Credit conditions and firm investment: Evidence from the MENA region
 - No 30 Michael Funke and Michael Paetz: Financial system reforms and China's monetary policy framework:
 - A DSGE-based assessment of initiatives and proposals
 - No 31 Zuzana Fungáčová, Pierre Pessarossi and Laurent Weill: Is bank competition detrimental to efficiency? Evidence from China
- 2013 No 1 Aaron Mehrotra: On the use of sterilisation bonds in emerging Asia
 - No 2 Zuzana Fungáčová, Rima Turk Ariss and Laurent Weill: Does excessive liquidity creation trigger bank failures?
 - No 3 Martin Gächter, Aleksandra Riedl and Doris Ritzberger-Grünwald: Business cycle convergence or decoupling? Economic adjustment in CESEE during the crisis
 - No 4 likka Korhonen and Anatoly Peresetsky: What determines stock market behavior in Russia and other emerging countries?
 - No 5 Andrew J. Filardo and Pierre L. Siklos: Prolonged reserves accumulation, credit booms, asset prices and monetary policy in Asia
 - No 6 Mustafa Disli, Koen Schoors and Jos Meir: Political connections and depositor discipline
 - No 7 Qiyue Xiong: The role of the bank lending channel and impacts of stricter capital requirements on the Chinese banking industry
 - No 8 Marek Dabrowski: Monetary policy regimes in CIS economies and their ability to provide price and financial stability
 - No 9 Rajeev K. Goel and Michael A. Nelson: Effectiveness of whistleblower laws in combating corruption
 - No 10 Yin-Wong Cheung and Rajeswari Sengupta: Impact of exchange rate movements on exports: An analysis of Indian non-financial sector firms
 - No 11 Martin Feldkircher, Roman Horvath and Marek Rusnak: Exchange market pressures during the financial crisis: A Bayesian model averaging evidence
 - No 12 Alicia Garcia-Herrero and Le Xia: China's RMB bilateral swap agreements: What explains the choice of countries?
 - No 13 Markus Eller, Jarko Fidrmuc and Zuzana Fungáčová: Fiscal policy and regional output volatility: Evidence from Russia
 - No 14 Hans Degryse, Liping Lu and Steven Ongena: Informal or formal financing? Or both? First evidence on the co-funding of Chinese firms
 - No 15 likka Korhonen and Anatoly Peresetsky: Extracting global stochastic trend from non-synchronous data
 - No 16 Roman Horvath, Jakub Seidler and Laurent Weill: How bank competition influence liquidity creation
 - No 17 Zuzana Fungáčová, Laura Solanko and Laurent Weill: Does bank competition influence the lending channel in the Euro area?
 - No 18 Konstantins Benkovskis and Julia Wörz: What drives the market share changes? Price versus non-price factors
 - No 19 Marcel P. Timmer and Ilya B. Voskoboynikov: Is mining fuelling long-run growth in Russia? Industry productivity growth trends since 1995
 - No 20 Iftekhar Hasan, Liang Song and Paul Wachtel: Institutional development and stock price synchronicity: Evidence from China

BOFIT Discussion Papers http://www.bof.fi/bofit_en • email: bofit@bof.fi ISSN 1456-6184, online